Author's response to reviews

Title: Expression of a novel carbonic anhydrase, CA XIII, in normal and neoplastic colorectal mucosa

Authors:

Laura Kummola (laura.kummola@uta.fi)
Jonna M Hamalainen (jonna.hamalainen@uta.fi)
Jyrki Kivela (jyrki.kivela@fimnet.fi)
Antti J Kivela (antti.j.kivela@fimnet.fi)
Juha Saarnio (juha.saarnio@oulu.fi)
Tuomo Karttunen (tuomo.karttunen@oulu.fi)
Seppo Parkkila (seppo.parkkila@uta.fi)

Version: 3 Date: 13 April 2005

Author's response to reviews:

The editor requested some stylistic changes to the revised version of the manuscript, and the modifications have been made according to the suggestions as follows:

Title- Can we ask you to revise the title to make it a single sentence, e.g. by using colons instead of a full stop.
- The title has been shortened to a single sentence.

Running title - Please remove the running title from the title page.
- The running title has been removed.

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals).
- The unnecessary capitals have been removed.

Abstract- the aim of the study should be incorporated into the Background section of the abstract and not under the Methods. Please remember to also update the abstract details on the submission page.
- The aim has been incorporated into the Background section of the abstract.

Introduction - Please rename this section 'Background'.
- The change has been made as suggested.

Materials & Methods - Please rename this section 'Methods'.
- The change has been made as suggested.

Conclusions; Competing interests; Authors' contributions; Acknowledgements- Please write the text under the corresponding section heading.
- The changes have been made as suggested.

References - The reference list should contain all authors' names, regardless of how many there are. The term "et al" should not be used.
- The references have been corrected as suggested.

Figure titles- Please provide a short title for each figure (15 words maximum).
- The short titles for each figure have been added.

Figures- The figures should be cropped to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping
- The figure 3 with white space around the image has been cropped.